**Author's response to reviews**

**Title:** Exercise Intervention During Pregnancy Can Be Used To Manage Weight Gain And Improve Pregnancy Outcomes In Women With Gestational Diabetes Mellitus

**Authors:**

Chen Wang (18518079870@163.com)
Weiwei Zhu (zhuweiwei@126.com)
Yumei Wei (weiyumei1982@126.com)
Hui Feng (fenghui1bt@163.com)
Rina Su (surina126014@126.com)
Huixia Yang (yanghuixia@bjmu.edu.cn)

**Version:** 4  **Date:** 14 September 2015

**Author's response to reviews:** see over
Dear editorial board of BMC Pregnancy and Childbirth,

Thanks very much for your kind work and consideration on publication of our paper (Exercise Intervention During Pregnancy Can Be Used To Manage Weight Gain And Improve Pregnancy Outcomes In Women With Gestational Diabetes Mellitus. ID 456446381727895).

We have studied comments carefully and have made correction which we hope meet with approval. Revised portion are marked in red in the paper. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Responds to the reviewer’s comments:

1. Please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state this in your manuscript. This applies to both the information collected through interviews and from medical records. If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

Response: We are very sorry for our negligence of stating that all participants in our study provided written informed consent. And the consent applies to both the information collected through interviews and from medical records. This has been clarified in the revised manuscript (page 4, line 93-94).

2. In the Authors Contributions section, please refer to the authors by their initials

Response: We are very sorry for our incorrect writing, and we have revised it in the manuscript (page 12, line 341-344).

We appreciate for Editors/Reviewers’ warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions.

Yours sincerely,

Chen Wang